Author's response to reviews

Title: Integrative network analysis identifies key genes and pathways in the progression of hepatitis C virus induced hepatocellular carcinoma

Authors:

Siyuan Zheng (siyuan.zheng@vanderbilt.edu)
William P Tansey (william.p.tansey@vanderbilt.edu)
Scott W Hiebert (scott.hiebert@vanderbilt.edu)
Zhongming Zhao (zhongming.zhao@vanderbilt.edu)

Version: 4 Date: 6 August 2011

Author's response to reviews:

We have added Methods subsection in Abstract and Conclusions section after the Discussion section. The web link in the Reference section should work now. Also, We have made some minor grammatical changes. Hope this is ready to go. Thanks,

Zhongming Zhao